Author's response to reviews

Title: Association Analysis of Chromosome 1 Migraine Candidate Genes

Authors:

Francesca Fernandez (f.fernandez@griffith.edu.au)
Robert P Curtain (robcurtain@hotmail.com)
Micky Ovcaric (m.ovcaric@griffith.edu.au)
John MacMillan (john_macmillan@health.qld.gov.au)
Lyn R Griffiths (l.griffiths@griffith.edu.au)

Version: 4 Date: 20 July 2007

Author's response to reviews: see over
Corrections to revised manuscript “Association Analysis of Chromosome 1 Migraine Candidate Genes”

Ref 1 in Reference list: Cephalgia, change to Cephalalgia

Ref 13 in line 10 of page 5 should be Ref [3]

Ref 23 in line 11 of page 10 should be Ref [24]

Within Table 3, in the first section entitled “ATP 1A4 SNP”:
6th line (MO), under CC should be 5 (13.6%) not 5 (13.5%)
8th line (Male), under TT should be 13 (39.3%) not 13 (39.4%)
9th line (Female), under TC should be 76 (43.1%) not 76 (43.2%)

Within Table 3, in the second section entitled “CAS Q1 SNP”:
6th line (MO), under C should be 42 (63.6) not 42 (63.4)

Within Table 3, in the third section entitled “KCN J9 SNP”:
5th line (MA), under TT should be 28 (14.2%) not 28 (14.3%)
6th line (MO), under N should be 74 not 66
6th line (MO), under T should be 26 (35.1%) not 26 (35.7%)
6th line (MO), under TT should be 4 (10.9%) not 0 (0%)

Other changes include addition of Methods section in Abstract and rearrangement of article sections to suit format requirements.